## **Author's Response To Reviewer Comments**

Clo<u>s</u>e

We would like to thank the editor and reviewers for their support throughout this process. We have cited the GigaDB dataset as requested and submit the updated manuscript. We would also like to thank the GigaDB database curation team for their time and diligence.

Clo<u>s</u>e